Author’s response to reviews

Title: Submicroscopic subtelomeric aberrations in Chinese patients with unexplained developmental delay/ mental retardation

Authors:

Ye Wu (dryewu@263.net)
Taoyun Ji (jitaoyun@163.com)
Jingmin Wang (jingmin66666@sina.com.cn)
Jing Xiao (wzcnyjl@126.com)
Huifang Wang (bei3210@sina.com.cn)
Jie Li (muzi19840105@sohu.com)
Zhijie Gao (gaozhijie20082008@163.com)
Yanling Yang (yanlingy@vip.sina.com)
Bin Cai (bcai@capitalbio.com)
Liwen Wang (wangliwen2@vip.sina.com)
Zhongshu Zhou (zhongshu_zhou@yahoo.com.cn)
Lili Tian (murasaki_ju@hotmail.com)
Xiaozhu Wang (wangzhu2242002@yahoo.com.cn)
Nan Zhong (genetomics@gmail.com)
Jiong Qin (qinjiong@263.net)
Xiru Wu (icnc@public3.bta.net.cn)
Yuwu Jiang (jiangyw@263.net)

Version: 4 Date: 29 April 2010

Author’s response to reviews:

Dear editor

I have done with the major (the orientation of Table 2 has been changed to be portrait), and the minor revisions for the manuscript.

Best wishes,
Sincerely yours,
Dr. Ye Wu